Author's Response To Reviewer Comments

Clo <u>s</u> e	
We sincerely thank Reviewer #2 for carefully examining our supplemental code during both revis stages. Their initial suggestions have certainly made the code more useful for our readers. Unfortunately, I am unable to reproduce the error described in the minor revision comments. I h even re-installed the most up-to-date version of ALDEx2 (version 1.16.0) to make sure that rece	ave
updates did not cause a new error. Perhaps the reviewer is using an old version of ALDEx2? Best regards, Thom Quinn	

Clo<u>s</u>e